Author’s response to reviews

Title: Optical puncture combined with balloon dilation PCNL vs. conventional puncture dilation PCNL for kidney stones without hydronephrosis: a retrospective study

Authors:
Mi Zhou (whateverzm@163.com)
Xiang He (hx.hzsycy@163.com)
Yuelong Zhang (zhngyuelong613@gmail.com)
Weiwen Yu (hfhyww@163.com)

Version: 1 Date: 12 Nov 2018

Author’s response to reviews:

Dear Editor:

1. Figure 2 and Figure 3 have been amended to delete individual's informations.

2. The section heading 'Materials and Methods' has been amended to 'Methods' in the main text of this manuscript.

Thx for your hard working!